

LJMU Research Online

Banach, M and Penson, PE

What have we learned about lipids and cardiovascular risk from PCSK9 inhibitor outcomes trials?

http://researchonline.ljmu.ac.uk/id/eprint/10210/

Article

Citation (please note it is advisable to refer to the publisher's version if you intend to cite from this work)

Banach, M and Penson, PE (2019) What have we learned about lipids and cardiovascular risk from PCSK9 inhibitor outcomes trials? Cardiovascular Research, 115 (3). pp. 26-31. ISSN 0008-6363

LJMU has developed LJMU Research Online for users to access the research output of the University more effectively. Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in LJMU Research Online to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.

The version presented here may differ from the published version or from the version of the record. Please see the repository URL above for details on accessing the published version and note that access may require a subscription.

For more information please contact researchonline@ljmu.ac.uk

http://researchonline.ljmu.ac.uk/

WHAT HAVE WE LEARNED ABOUT LIPIDS AND CARDIOVASCULAR RISK FROM PCSK9 INHIBITOR OUTCOMES TRIALS?

Maciej Banach^{1-3#} and Peter E. Penson⁴

¹ Department of Hypertension, Chair of Nephrology and Hypertension, Medical University of Lodz, Lodz, Poland; ²Polish Mother's Memorial Hospital Research Institute (PMMHRI), Lodz, Poland, ³Cardiovascular Research Centre, University of Zielona Gora, Zielona Gora, Poland. ⁴School of Pharmacy & Biomolecular Sciences, Liverpool John Moores University, Liverpool, UK;

Keywords: alirocumab, evolocumab, familial hypercholesterolaemia, PCSK9,

#Address for correspondence:

Prof. Maciej Banach, MD, PhD, FNLA, FAHA, FESC, FASA, Head, Department of Hypertension, WAM University Hospital in Lodz, Medical University of Lodz, Zeromskiego 113; 90-549 Lodz, Poland. Phone: +48426393771; Fax: +48 42 639 37 71; E-mail: maciejbanach77@gmail.com

ABBREVIATIONS:

ACC	American College of Cardiology		
ACS	Acute Coronary Syndromes		
АНА	American Heart Association		
ASCVD	Atherosclerotic cardiovascular disease		
CANTOS	Canakinumab Antiinflammatory Thrombosis Outcome Study		
EBBINGHAUS	Evaluating PCSK9 Binding Antibody Influence on Cognitive		
	Health in High Cardiovascular Risk Subjects		
FH	Familial hypercholesterolaemia		
FOURIER	Further Cardiovascular Outcomes Research with PCSK9		
	Inhibition in Subjects with Elevated Risk		
LDL-C	Low-density lipoprotein cholesterol		
Lp(a)	Lipoprotein(a)		
ODYSSEY Outcomes	Evaluation of Cardiovascular Outcomes After an Acute		
	Coronary Syndrome During Treatment With Alirocumab		
PCSK9:	Proprotein convertase subtilisin/kexin type 9		
SAMS	Statin-associated muscle symptoms		
siRNA	Small interfering ribonucleic acid		

1. Introduction

The role of proprotein convertase subtilisin/kexin type 9 (PCSK9) in the regulation of the LDL-receptor and plasma cholesterol was identified in 2003¹. This seminal discovery led to the rapid exploitation of PCSK9 as a drug target. Fifteen years later, investigators have completed two large outcomes trials of monoclonal antibody inhibitors of PCSK9 (PCSK9Is). Evolocumab was evaluated in The Further Cardiovascular Outcomes Research with PCSK9 Inhibition in Subjects with Elevated Risk (FOURIER)², and alirocumab was tested in the Evaluation of Cardiovascular Outcomes After an Acute Coronary Syndrome During Treatment With Alirocumab (ODYSSEY Outcomes)³ trials (**Table 1**). Treatment of patients with these agents has been remarkably successful in achieving lower plasma concentrations of LDL-C than had ever previously been attainable with lipid lowering therapy. Therefore, in addition to the 'headline' event-rate reduction results of these trials, which will certainly inform clinical practice, scrutiny of the trial data can yield more general information about lipid-lowering therapy. This commentary will briefly discuss what we have learned from outcome trials of evolocumab and alirocumab, and what the future holds for these agents.

2. What have we learned from PCSK9I outcomes trials, and how has practice changed?

The addition of alirocumab and evolocumab to lipid-lowering therapy with statins (with or without ezetimbe), has enabled the effective management of patient groups for whom achievement and maintenance of lipid-targets were previously extremely difficult. In particular the very large reductions in LDL-C elicited by PCSK9 inhibitors has been used to manage patients with familial hypercholesterolaemia (FH). Using these agents, it has been possible to meet LDL-targets for most of this population, and additionally for some of them (even for 93%) it is now possible to entirely replace LDL-C apheresis (or to reduce the frequency of procedures) using PCSK9 inhibitor therapy. In several European countries, FH patients are the

only group for whom reimbursement for PCSK9 inhibitors is allowed (**Figure 1**). The availability of an effective therapeutic approach to FH has in fact led to a revival in interest in this condition⁴ in many countries in which knowledge of this condition (and other genetically based lipid disorders) was very poor. Hopefully this will lead to redoubled efforts to identify undiagnosed FH patients.

Statin intolerance, in particular statin-associated muscle symptoms (SAMS) is an important cause of statin non-adherence, discontinuation and suboptimal lipid management ^{5,6}. Inhibitors of PCSK9 have also prompted discussion on the definition of statin intolerance and its effective diagnosis⁷. Additionally these agents have shown promise in the effective management of plasma lipid concentrations in statin intolerant patients, for whom there was previously no effective therapy⁵⁻⁷. The GAUSS-3 trial found that in patients with confirmed statin-intolerance, evolocumab was associated with a 53% reduction in LDL-C at 24 weeks. Muscle symptoms were reported in only 21% of evolocumab-treated patients⁸. Similarly, in the ODYSSEY-alternative trial, alirocumab reduced mean LDL-C by 45% at 24 weeks⁹. In both cases PCSK9 inhibition was superior to ezetimibe. The demonstration of the concept of effective lipid-lowering in statin intolerant patients has led to trials investigating alternative lipid-lowering therapies such as bempedoic acid and to look for the other alternative therapies that might reduce LDL-C in patients with high cardiovascular (CV) risk^{10,11}.

The extremely low plasma concentrations of LDL-C achieved by treatment with PCSK9 inhibitors has renewed focus on residual (i.e. non-LDL-C–mediated) risk for cardiovascular disease (CVD). This is timely in light of the recent demonstrations of the importance of inflammatory risk factors for cardiovascular disease in the Canakinumab Antiinflammatory Thrombosis Outcome Study (CANTOS)¹² and Reasons for Geographic and Racial Differences in Stroke (REGARDS) study¹³. Additionally, the importance of lipoprotein(a) (Lp(a)) in mediating CV risk has received renewed attention. It is worth mentioning that purely on the basis of data with PCSK9 inhibitors, physicians in many countries worldwide have started

(re)measuring Lp(a). This highly atherogenic and pro-thrombotic lipoprotein is resistant to many lipid-lowering therapies¹⁴, but appears to be effectively reduced by PCSK9 inhibitors¹⁵.

Importantly, FOURIER² and ODYSSEY-Outcomes³ have demonstrated conclusively that additional lipid-lowering therapy can be very beneficial in reducing cardiac outcomes in patients already taking optimal statin therapy. Previously, the Improved Reduction of Outcomes: Vytorin Efficacy International Trial (IMPROVE-IT) had demonstrated a reduction of a composite endpoint when ezetimibe was added to statin therapy¹⁶, however ODYSSEY-Outcomes additionally demonstrated an important reduction in all-cause mortality with alirocumab ³

Furthermore, ODYSSEY-Outcomes³ and FOURIER² have added further evidence to support the 'lower is better for longer' paradigm with respect to LDL-C. In alirocumab-treated patients mean LDL-C levels at 4 months, 12 months, and 48 months were 0.98 mmol/l (38 mg/dl), 1.1 mmol/l (42 mg/dl), and 1.4 mmol/l (53 mg/dl), respectively with the composite primary outcome (death from coronary heart disease, nonfatal myocardial infarction, fatal or nonfatal ischemic stroke, or unstable angina requiring hospitalization) reduced by 15% (95%Cl: 0.78-0.93) (with absolute risk reduction [aRR] 2% - number needed to treat [NNT]: 50]³. Patients treated with evolocumab achieved mean LDL-C of 2.4 mmol/l (30 mg/dl) and this was associated with a 15% reduction (95%Cl: 0.79-0.92; aRR=2%, NNT=50) in the primary outcome (a composite of cardiovascular death, myocardial infarction, stroke, hospitalisation for unstable angina or coronary revascularisation)². A secondary analysis of the data revealed that the benefits persisted at LDL-C levels <0.5 mmol/L (20 mg/dL)¹⁷. Importantly, these benefits do not appear to be associated with excessive adverse effects and reassuringly, the rigorously conducted Evaluating PCSK9 Binding Antibody Influence on Cognitive Health in High Cardiovascular Risk Subjects (EBBINGHAUS) study found no evidence of cognitive impairment over a median of 19 months of evolocumab therapy¹⁸.

How should PCSKIs be used in high-risk patients?

The high acquisition cost of PCSK9 inhibitors, and the fact that high-risk individuals have been included in clinical trials has led to a discussion about the most appropriate method to stratify risk in the secondary prevention of CVD. Risk stratification is essential in order to select the patients who are most likely to benefit from treatments, and consequently to improve the cost-effectiveness of these medicines in measures such as Quality Adjusted Life Years (QALYs) and other techniques used economic evaluations of medicines. Based on the previous analyses PCSK9 inhibitor therapy should be considered in all high-risk patient groups where the number needed to treat (NNT) is less than 30^{19} . Using the available data and sub-analyses of PCSK9 trials such patient groups have been identified by the International Lipid Expert Panel (ILEP)²⁰ (Table 2), however further development of risk stratification algorithms would be extremely beneficial in deciding whether to initiate therapy with statin monotherapy, or to use combination therapy with statins and ezetimibe or a PCSK9 inhibitor from the outset. A risk stratification tool has previously been developed to assist therapeutic decision-making with respect to the initiation of the thrombin receptor antagonist vorapaxar in patients with ischemic heart disease²¹. A similar scheme for decision making with respect to PCSK9 inhibitors would likely include clinical details in addition to baseline lipid levels¹⁹ and coronary artery calcium scores^{22,23} to enable optimal decision making with respect to initial treatment. Based on these analyses we should consider changes in risk stratification for secondary prevention patients and introducing the group of extremely high risk patients to finally give clear answers as to which patients should be treated with the highest statin doses (80 mg atorvastatin and 40 mg rosuvastatin instead of the recommended ranges of 40-80 and 20-40 mg for atorvastatin and rosuvastatin, respectively), which patients should initiate treatment with combination therapy, and which ones might benefit the most from new therapies²⁰. Furthermore, recently published findings and ongoing studies may potentially expand the scope of PCSK9I therapy to include

new patient groups including pregnancy, primary prevention and comorbidities such as chronic kidney disease, and diabetes ^{20,24}.

What does the future hold for PCSK9 inhibitors?

Recently published data from ODYSSEY-Outcomes³ and FOURIER² suggests that evolocumab and alirocumab are extremely effective in lowering atherogenic lipoproteins, and consequently in preventing major adverse CV events. Availability of PCSK9 inhibitors to patients is determined by local policies on reimbursement. Such policies vary widely between countries, with no availability still in some countries, or with the reimbursement only for limited group of patients with FH in others (Figure 1). Cost-benefit analyses are likely to move in favour of PCSK9 inhibitors because of recent price reductions in these agents. In October 2018, Amgen announced that it was lowering the price of Repatha® (evolocumab) by, by approximately 60% from \$14,000 to \$5850 a year. This followed an announcement in May 2018 that Regeneron and Sanofi, the manufacturers of Praluent® (alirocumab) would reduce the price of this agent for a large number of patients. Physician familiarity with, and use of PCSK9Is is likely to increase following the publication of the recent American College of Cardiology and American Heart Association (ACC/AHA) Guideline on the Management of Blood Cholesterol. The guideline makes a class IIa recommendation for patients with multiple CV risk factors following acute coronary syndromes (ACS); but surprisingly a class IIb recommendation for patients with FH²⁵.

With excellent clinical trial results and improving affordability, PCSK9-inibitors are likely to be a rational choice of lipid-lowering therapy in an increasing number of very high and extremely high-risk patients. In parallel with the rapid clinical developments, PCSK9 has continued to be remarkably fruitful field of study for basic scientists. As an alternative to monoclonal antibody inhibitors to PCSK9, the ORION series of clinical trials have demonstrated safety and lipid-lowering efficacy of inclisiran, a small interfering RNA (siRNA) which targets the messenger RNA for PCSK9²⁶. Recent preclinical studies have shown promising results for an anti-PCSK9 antisense oligonucleotide, which reduces serum PCSK9 in a rat model ²⁷.

The remarkably rapid progress from the identification of PCSK9 as a drug target to the development, clinical evaluation and regulatory approval of multiple PCSK9Is is almost unique in modern medicine. This success is clearly due to close collaboration between basic scientists and clinical researchers in high-quality translational research projects. It is hoped that continued collaborative research will lead to more cost-effective therapies so that the remarkable benefits of PCSK9 inhibition can be extended to as many patients as possible.

Declaration of Interest:

Maciej Banach: speakers bureau: Abbott/Mylan, Abbott Vascular, Actavis, Akcea, Amgen, Biofarm, KRKA, MSD, Sanofi-Aventis, Servier and Valeant; consultant to Abbott Vascular, Akcea, Amgen, Daichii Sankyo, Esperion, Lilly, MSD, Polfarmex, Resverlogix, Sanofi-Aventis; Grants from Sanofi and Valeant; *Peter Penson* owns four shares in AstraZeneca PLC and has received speaker's fees from Amgen Inc.

REFERENCES:

1. Abifadel M, Varret M, Rabes JP, Allard D, Ouguerram K, Devillers M, Cruaud C, Benjannet S, Wickham L, Erlich D, Derre A, Villeger L, Farnier M, Beucler I, Bruckert E, Chambaz J, Chanu B, Lecerf JM, Luc G, Moulin P, Weissenbach J, Prat A, Krempf M, Junien C, Seidah NG, Boileau C. Mutations in PCSK9 cause autosomal dominant hypercholesterolemia. Nat Genet 2003;**34**(2):154-6.

2. Sabatine MS, Giugliano RP, Keech AC, Honarpour N, Wiviott SD, Murphy SA, Kuder JF, Wang H, Liu T, Wasserman SM, Sever PS, Pedersen TR, Fourier Steering Committee

Investigators. Evolocumab and Clinical Outcomes in Patients with Cardiovascular Disease. N Engl J Med 2017;**376**(18):1713-1722.

3. Schwartz GG, Steg PG, Szarek M, Bhatt DL, Bittner VA, Diaz R, Edelberg JM, Goodman SG, Hanotin C, Harrington RA, Jukema JW, Lecorps G, Mahaffey KW, Moryusef A, Pordy R, Quintero K, Roe MT, Sasiela WJ, Tamby JF, Tricoci P, White HD, Zeiher AM, Odyssey Outcomes Committees Investigators. Alirocumab and Cardiovascular Outcomes after Acute Coronary Syndrome. N Engl J Med 2018; doi: 10.1056/NEJMoa1801174.

4. Vallejo-Vaz AJ, De Marco M, Stevens CAT, Akram A, Freiberger T, Hovingh GK, Kastelein JJP, Mata P, Raal FJ, Santos RD, Soran H, Watts GF, Abifadel M, Aguilar-Salinas CA, Al-Khnifsawi M, AlKindi FA, Alnouri F, Alonso R, Al-Rasadi K, Al-Sarraf A, Ashavaid TF, Binder CJ, Bogsrud MP, Bourbon M, Bruckert E, Chlebus K, Corral P, Descamps O, Durst R, Ezhov M, Fras Z, Genest J, Groselj U, Harada-Shiba M, Kayikcioglu M, Lalic K, Lam CSP, Latkovskis G, Laufs U, Liberopoulos E, Lin J, Maher V, Majano N, Marais AD, Marz W, Mirrakhimov E, Miserez AR, Mitchenko O, Nawawi HM, Nordestgaard BG, Paragh G, Petrulioniene Z, Pojskic B, Postadzhiyan A, Reda A, Reiner Z, Sadoh WE, Sahebkar A, Shehab A, Shek AB, Stoll M, Su TC, Subramaniam T, Susekov AV, Symeonides P, Tilney M, Tomlinson B, Truong TH, Tselepis AD, Tybjaerg-Hansen A, Vazquez-Cardenas A, Viigimaa M, Vohnout B, Widen E, Yamashita S, Banach M, Gaita D, Jiang L, Nilsson L, Santos LE, Schunkert H, Tokgozoglu L, Car J, Catapano AL, Ray KK, E. A. S. Familial Hypercholesterolaemia Studies Collaboration Investigators. Overview of the current status of familial hypercholesterolaemia care in over 60 countries - The EAS Familial Hypercholesterolaemia Studies Collaboration (FHSC). Atherosclerosis 2018;277:234-255.

Penson PE, Mancini GBJ, Toth PP, Martin SS, Watts GF, Sahebkar A, Mikhailidis DP,
 Banach M, Lipid, Blood Pressure Meta-Analysis Collaboration G, International Lipid Expert
 P. Introducing the 'Drucebo' effect in statin therapy: a systematic review of studies comparing

reported rates of statin-associated muscle symptoms, under blinded and open-label conditions. J Cachexia Sarcopenia Muscle 2018; doi: 10.1002/jcsm.12344

6. Rosenson RS, Baker S, Banach M, Borow KM, Braun LT, Bruckert E, Brunham LR, Catapano AL, Elam MB, Mancini GBJ, Moriarty PM, Morris PB, Muntner P, Ray KK, Stroes ES, Taylor BA, Taylor VH, Watts GF, Thompson PD. Optimizing Cholesterol Treatment in Patients With Muscle Complaints. J Am Coll Cardiol. 2017;**70**:1290-1301.

7. Banach M, Rizzo M, Toth PP, Farnier M, Davidson MH, Al-Rasadi K, Aronow WS, Athyros V, Djuric DM, Ezhov MV, Greenfield RS, Hovingh GK, Kostner K, Serban C, Lighezan D, Fras Z, Moriarty PM, Muntner P, Goudev A, Ceska R, Nicholls SJ, Broncel M, Nikolic D, Pella D, Puri R, Rysz J, Wong ND, Bajnok L, Jones SR, Ray KK, Mikhailidis DP. Statin intolerance - an attempt at a unified definition. Position paper from an International Lipid Expert Panel. Arch Med Sci. 2015;**11**:1-23.

8. Nissen SE, Stroes E, Dent-Acosta RE, Rosenson RS, Lehman SJ, Sattar N, Preiss D, Bruckert E, Ceska R, Lepor N, Ballantyne CM, Gouni-Berthold I, Elliott M, Brennan DM, Wasserman SM, Somaratne R, Scott R, Stein EA, Investigators G-. Efficacy and Tolerability of Evolocumab vs Ezetimibe in Patients With Muscle-Related Statin Intolerance: The GAUSS-3 Randomized Clinical Trial. JAMA 2016;**315**(15):1580-90.

9. Moriarty PM, Thompson PD, Cannon CP, Guyton JR, Bergeron J, Zieve FJ, Bruckert E, Jacobson TA, Kopecky SL, Baccara-Dinet MT, Du Y, Pordy R, Gipe DA, Odyssey Alternative Investigators. Efficacy and safety of alirocumab vs ezetimibe in statin-intolerant patients, with a statin rechallenge arm: The ODYSSEY ALTERNATIVE randomized trial. J Clin Lipidol 2015;**9**(6):758-769.

10. Penson P, McGowan M, Banach M. Evaluating bempedoic acid for the treatment of hyperlipidaemia. Expert Opin Investig Drugs 2017;**26**(2):251-259.

11. Banach M, Patti AM, Giglio RV, Cicero AFG, Atanasov AG, Bajraktari G, Bruckert E, Descamps O, Djuric DM, Ezhov M, Fras Z, von Haehling S, Katsiki N, Langlois M,

Latkovskis G, Mancini GBJ, Mikhailidis DP, Mitchenko O, Moriarty PM, Muntner P, Nikolic D, Panagiotakos DB, Paragh G, Paulweber B, Pella D, Pitsavos C, Reiner Ž, Rosano GMC, Rosenson RS, Rysz J, Sahebkar A, Serban MC, Vinereanu D, Vrablík M, Watts GF, Wong ND, Rizzo M; International Lipid Expert Panel (ILEP). The Role of Nutraceuticals in Statin Intolerant Patients. J Am Coll Cardiol. 2018;**72**(1):96-118.

12. Ridker PM, Everett BM, Thuren T, MacFadyen JG, Chang WH, Ballantyne C, Fonseca F, Nicolau J, Koenig W, Anker SD, Kastelein JJP, Cornel JH, Pais P, Pella D, Genest J, Cifkova R, Lorenzatti A, Forster T, Kobalava Z, Vida-Simiti L, Flather M, Shimokawa H, Ogawa H, Dellborg M, Rossi PRF, Troquay RPT, Libby P, Glynn RJ, Cantos Trial Group. Antiinflammatory Therapy with Canakinumab for Atherosclerotic Disease. N Engl J Med 2017;**377**(12):1119-1131.

13. Penson PE, Long DL, Howard G, Toth PP, Muntner P, Howard VJ, Safford MM, Jones SR, Martin SS, Mazidi M, Catapano AL, Banach M. Associations between very low concentrations of low density lipoprotein cholesterol, high sensitivity C-reactive protein, and health outcomes in the Reasons for Geographical and Racial Differences in Stroke (REGARDS) study. Eur Heart J 2018;**39**(40):3641-3653.

14. Kotani K, Serban MC, Penson P, Lippi G, Banach M. Evidence-based assessment of lipoprotein(a) as a risk biomarker for cardiovascular diseases - Some answers and still many questions. Crit Rev Clin Lab Sci 2016;**53**(6):370-8.

15. Cao YX, Liu HH, Li S, Li JJ. A Meta-Analysis of the Effect of PCSK9-Monoclonal Antibodies on Circulating Lipoprotein (a) Levels. Am J Cardiovasc Drugs 2018; doi: 10.1007/s40256-018-0303-2.

16. Cannon CP, Blazing MA, Giugliano RP, McCagg A, White JA, Theroux P, Darius H, Lewis BS, Ophuis TO, Jukema JW, De Ferrari GM, Ruzyllo W, De Lucca P, Im K, Bohula EA, Reist C, Wiviott SD, Tershakovec AM, Musliner TA, Braunwald E, Califf RM, Improve-

It Investigators. Ezetimibe Added to Statin Therapy after Acute Coronary Syndromes. N Engl J Med 2015;**372**(25):2387-97.

17. Giugliano RP, Pedersen TR, Park J-G, De Ferrari GM, Gaciong ZA, Ceska R, Toth K, Gouni-Berthold I, Lopez-Miranda J, Schiele F, Mach F, Ott BR, Kanevsky E, Pineda AL, Somaratne R, Wasserman SM, Keech AC, Sever PS, Sabatine MS. Clinical efficacy and safety of achieving very low LDL-cholesterol concentrations with the PCSK9 inhibitor evolocumab: a prespecified secondary analysis of the FOURIER trial. The Lancet 2017;**390**(10106):1962-1971.

 Giugliano RP, Mach F, Zavitz K, Kurtz C, Im K, Kanevsky E, Schneider J, Wang H, Keech A, Pedersen TR, Sabatine MS, Sever PS, Robinson JG, Honarpour N, Wasserman SM, Ott BR, Ebbinghaus Investigators. Cognitive Function in a Randomized Trial of Evolocumab. N Engl J Med 2017;**377**(7):633-643.

19. Robinson JG, Huijgen R, Ray K, Persons J, Kastelein JJ, Pencina MJ. Determining When to Add Nonstatin Therapy: A Quantitative Approach. J Am Coll Cardiol 2016;68(22):2412-2421.

20. International Lipid Expert Panel (ILEP) Position Paper on PCSK9 inihibitors management and reimbursment rules. In preparation 2018.

21. Bohula EA, Bonaca MP, Braunwald E, Aylward PE, Corbalan R, De Ferrari GM, He P, Lewis BS, Merlini PA, Murphy SA, Sabatine MS, Scirica BM, Morrow DA. Atherothrombotic Risk Stratification and the Efficacy and Safety of Vorapaxar in Patients With Stable Ischemic Heart Disease and Previous Myocardial Infarction. Circulation 2016;**134**(4):304-313.

22. Yano Y, O'Donnell CJ, Kuller L, Kavousi M, Erbel R, Ning H, D'Agostino R, Newman AB, Nasir K, Hofman A, Lehmann N, Dhana K, Blankstein R, Hoffmann U, Mohlenkamp S, Massaro JM, Mahabadi AA, Lima JAC, Ikram MA, Jockel KH, Franco OH, Liu K, Lloyd-Jones D, Greenland P. Association of Coronary Artery Calcium Score vs Age With

Cardiovascular Risk in Older Adults: An Analysis of Pooled Population-Based Studies. JAMA Cardiol 2017;**2**(9):986-994.

23. Polonsky TS, McClelland RL, Jorgensen NW, Bild DE, Burke GL, Guerci AD, Greenland P. Coronary artery calcium score and risk classification for coronary heart disease prediction. JAMA 2010;**303**(16):1610-6.

24. Bittner VA, Giugliano RP, Brinton EA, Guyton JR. PCSK9 inhibitors for prevention of atherosclerotic cardiovascular disease. J Clin Lipidol 2018;**12**(4):835-843.

25. Grundy SM, Stone NJ, Bailey AL, Beam C, Birtcher KK, Blumenthal RS, Braun LT, de Ferranti S, Faiella-Tommasino J, Forman DE, Goldberg R, Heidenreich PA, Hlatky MA, Jones DW, Lloyd-Jones D, Lopez-Pajares N, Ndumele CE, Orringer CE, Peralta CA, Saseen JJ. Smith SC. Jr.. Sperling L, Virani SS. Yeboah J. 2018 AHA/ACC/AACVPR/AAPA/ABC/ACPM/ADA/AGS/APhA/ASPC/NLA/PCNA Guideline on the Management of Blood Cholesterol: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. J Am Coll Cardiol 2018.

26. Ray KK, Stoekenbroek RM, Kallend D, Leiter LA, Landmesser U, Wright RS, Wijngaard P, Kastelein JJP. Effect of an siRNA Therapeutic Targeting PCSK9 on Atherogenic Lipoproteins. Circulation 2018;**138**(13):1304-1316.

27 Harada-Shiba M, Wada F, Tachibana K, Kobayashi T. Development of Antisense Drug Targeting PCSK9 Circulation 2010; **136** (S1) A18674.

Figure 1. The information on reimbursement of PCSK9 inhibitors in Europe (status as of 1st Nov. 2018).



Table 1: Comparison of FOURIER and ODYSSEY-Outcomes trials

	FOURIER ²	ODYSSEY-Outcomes ³			
Characteristics of study and participants					
Publication year	2017	2018			
Trial design	Randomised double-blind placebo-controlled trial	Randomised double-blind placebo-controlled			
		trial			
Drug regiments	Evolocumab subcutaneously either 140mg every 2	Alirocumab (dose adjusted to meet LDL-C			
	weeks or 420mg monthly	target of 25 to 50 mg/dl)			
Mean follow-up	2.2 years	2.8 years			
Number of participants	27,564	18.924			
Mean age of participants	62.5 ± 9.1 (evolocumab)	58.5 ± 9.3 (alirocumab)			
	62.5 ± 8.9 (placebo)	58.6 ± 9.4 (placebo)			
% of male participants	75.4% evolocumab	74.7 (alirocumab)			
	75.5% placebo	74.9 (placebo)			
Main inclusion criteria	Age >40 <85	Age >40			
	Clinically relevant ASCVD	Hospitalized with ACS 1-12 months before			
	Additional 'high risk' characteristics	randomization			
	Fasting LDL-C \geq 70mg/dl or non-HDL-C \geq	$LDL-C \ge 70 \text{ mg/dl}$			
	100mg/dl on optimal therapy	Non HDL-C \geq 100mg/dl or apolipoprotein B			
		$\geq 80 \text{ mg/dl}.$			
Background therapy	High intensity statin (69.3%)	High intensity atorvastatin/rosuvastatin			
	Moderate intensity statin (30.4%)	(88.9%)			
	Low intensity/ no data (0.03%)	Low/ moderate intensity			
	Ezetimibe (5.3%)	atorvastatin/rosuvastatin (8.5%)			
		Other statin (0.25%)			
		No Statin (2.4%)			

		Ezetimibe (2.9%)				
Endpoints and Results						
Primary endpoint and hazard ratio	Composite of cardiovascular death, myocardial	Composite of death from coronary heart				
	infarction, stroke, hospitalisation for unstable angina	disease, nonfatal myocardial infarction, fatal				
	or coronary revascularisation. HR (95%CI) v	or nonfatal ischemic stroke, or unstable				
	placebo 0.85 (0.79-0.92)	angina requiring hospitalization. HR (95%CI)				
		v placebo 0.85 (0.78-0.93)				

Table 2: Selected groups of patients that might benefit most from the use of PCSK9 inhibitors based on the most recent data summarized by the International Lipid Expert Panel (ILEP)²⁰.

Patient group	Background	Biomarkers	NNT
	therapy		
Patients with early ACS (up to 1-12 months)	Optimal	Persistent LDL≥100 mg/dl	29
	treatment		
Patient with fresh ACS (up to 1-12 months)	Optimal	Persistent LDL 270 mg/dl + diabetes mellitus and/or baseline Lp(a)>	30
	treatment	60 mg/dl	
Patients with very high cardiovascular risk	Optimal	Persistent LDL 270 mg/dl + diabetes and/or baseline CRP> 3 mg/dl	<30
(after ACS)	treatment		
Patient in the group of very high	Optimal	Persistent LDL 270 mg/dl + with concomitant PAD	29
cardiovascular risk (after ACS) on	treatment		
Patients with very high cardiovascular risk	Optimal	Persistent LDL \geq 70 mg/dl + \geq 2 previous ACS and initially with	<30
(after ACS)	treatment	diabetes / Lp(a)>60 mg/dl / CRP> 3 mg/l	
Patients with very high cardiovascular risk	Optimal	Persistent LDL≥70 mg/dl + with multivessel disease	29
(after ACS)	treatment		